Author’s response to reviews

Title: Lack of efficacy of pomegranate supplementation for glucose management, insulin levels and sensitivity: evidence from a systematic review and meta-analysis

Authors:

Haohai Huang (haohaihuang@hotmail.com)
Dan Liao (26312526@qq.com)
Guangzhao Chen (gzchen90@126.com)
Honglang Chen (915475709@qq.com)
Yongkun Zhu (379805596@qq.com)

Version: 2 Date: 15 Sep 2017

Author’s response to reviews:

Sep. 15, 2016

Dear Prof. Kirkpatrick,

On behalf of my co-authors, I thank you very much for giving us an opportunity to revise our manuscript entitled "Lack of efficacy of pomegranate supplementation on glucose management, insulin levels and sensitivity: evidence from a systematic review and meta-analysis (NUTJ-D-17-00177R1)". We also appreciate the reviewers very much for their positive and constructive comments and suggestions on our manuscript.

We have studied reviewers’ comments carefully and tried our best to make revisions which are marked in yellow in the revised manuscript. Attached please find the revised version, which we would like to re-submit for your kind consideration. In addition, point-by-point responses to the comments are listed as following:

Responds to your comments:

Reviewer #1: My comments have been adequately addressed.

Nevertheless, the manuscript still needs some language editing.
Response: Thanks for your useful suggestions. The revised manuscript has been edited and proofread by the Wiley Editing services company. Attached please find the revised version, which I would like to re-submit for your kind consideration. You should be able to see all of our changes when you open the file. All changes are marked in yellow. I hope that the revised manuscript is now acceptable for publication in the journal.

Once again, thank you very much for your comments and suggestions. Please also advise if you need any additional information for the revised manuscript. Thank very much for your consideration, and look forward to hearing from you soon.

Yours sincerely,

Haohai Huang, M.D.

Department of Clinical Pharmacy, Dongguan Third People's Hospital, Affiliated Dongguan Shilong People’s Hospital of Southern Medical University